Author's response to reviews

Title: Teenage drinking, alcohol availability and pricing: a cross-sectional study of risk and protective factors for alcohol-related harms in school children

Authors:

Mark A Bellis (m.a.bellis@ljmu.ac.uk)
Penelope A Phillips-Howard (p.phillips-howard@ljmu.ac.uk)
Karen Hughes (k.e.hughes@ljmu.ac.uk)
Sara Hughes (s.k.hughes@ljmu.ac.uk)
Penny A Cook (p.a.cook@ljmu.ac.uk)
Michela Morleo (m.j.morleo@ljmu.ac.uk)
Kerin Hannon (k.hannon@ljmu.ac.uk)
Linda Smallthwaite (Linda.Smallthwaite@halton.gov.uk)
Lisa Jones (l.jones1@ljmu.ac.uk)

Version: 5 Date: 21 September 2009

Author's response to reviews:

Dear Editor,

We are delighted that the paper: Teenage drinking, alcohol availability and pricing: a cross-sectional study of risk and protective factors for alcohol-related harms in school children has now been accepted for publication in BMC Public Health. We have addressed all the formatting requirements. However, if you require anything further please do not hesitate to contact me.

Best wishes,

Mark